Author's response to reviews

Title: Association between metabolic abnormalities and HBV related hepatocellular carcinoma in Chinese: A cross-sectional study

Authors:

Jinyan Zhao (jingy1213@163.com)
Yunpeng Zhao (jane124600@163.com)
Hao Wang (whowardwh@163.com)
Xing Gu (Sindygux@gmail.com)
Jun Ji (jillfly1217@163.com)
Chunfang Gao (gaocf1115@163.com)

Version: 6 Date: 13 May 2011

Author's response to reviews:

May 9, 2011

Hello, Dear Editors:

Thank you very much for accepting our manuscript. In the revised manuscript, we have made the following changes that you referred to point by point. We have added the section titled “Additional files” at the end of the manuscript (after the tables) #the format of which was the same as the example you given; The figure was cropped as closely as possible to minimise white space around the image; The reference list showed all the authors’ names for each citation; The acknowledgments has been moved to after the competing interests; All coloured has been formatted to black.

We are looking forward to hearing from you soon!

Chunfang Gao, MD, PhD (corresponding author)
Professor and Director of Department of Laboratory Medicine
Eastern Hepatobiliary Surgical Hospital
Second Military Military Medical University
Shanghai, China